Author's response to reviews

Title: The influence of psychiatric screening in healthy populations selection: a new study and meta-analysis of functional 5-HTTLPR and rs25531 polymorphisms and anxiety-related personality traits

Authors: 

Alessandra Minelli (aminelli@fatebenefratelli.it)  
Cristian Bonvicini (cbonvicini@fatebenefratelli.it)  
Catia Scassellati (cscassellati@fatebenefratelli.it)  
Riccardo Sartori (riccardo.sartori@univr.it)  
Massimo Gennarelli (gennarelli@fatebenefratelli.it)

Version: 8 Date: 21 February 2011

Author's response to reviews:

Dear Editor

sorry for the mistakes, we made the needed changes.

Regards